Author's response to reviews

Title: Clinicopathologic and gene expression parameters predicts liver cancer prognosis

Authors:

Ke Hao (ke_hao@merck.com)
John Lamb (John.Lamb@pfizer.com)
Chunsheng Zhang (chunsheng_zhang@merck.com)
Tao Xie (Tao.Xie@pfizer.com)
Kai Wang (Kai.Wang4@pfizer.com)
Bin Zhang (binzhang.ucla@gmail.com)
Eugene Chudin (eugene.chudin@gmail.com)
Nikki P Lee (nikkilee@hkucc.hku.hk)
Mao Mao (Mao.Mao@pfizer.com)
Hua Zhong (judy.zhong@nyumc.org)
Danielle Greenawalt (Danielle_Greenawalt@merck.com)
Mark D Ferguson (mark_Ferguson@merck.com)
Irene O Ng (iolng@hku.hk)
Pak C Sham (pcsham@hku.hk)
Ronnie T Poon (poontp@hku.hk)
Cliona Molony (cliona_molony@merck.com)
Eric E Schadt (eric.schadt@gmail.com)
Hongyue Dai (hongyue_dai@merck.com)
John M Luk (luk_john_moonching@nuhs.edu.sg)

Version: 6 Date: 7 November 2011

Author's response to reviews: see over
Title: Clinicopathologic and Gene Expression Parameters Predicts Liver Cancer Prognosis
Journal: BMC Cancer
MS ID: 2450090205286408

Dear the editor,

Thanks for you comments and we have made formatting changes according to your suggestions

--------------------------------
Major revisions (we require the author to make these changes)

Additional files: Please cite the additional files in numerical order. Currently, the files are not in numerical order.
Additional files: Please do not use ranges when listing additional files. Please list all the files.

In the revised version, additional files are cited in ascending numerical order: We do not use ranges when listing additional files.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Structure: Change introduction to background.

We changed introduction to background

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We carefully checked the typographical errors in the manuscript.

Thank you,

--Ke